Author's response to reviews

Title: Hepatoprotection with chloroform extract of Launaea procumbens against CCl4 induced injuries in rat

Authors:

Rahmat Ali Khan i (rahmatgul_81@yahoo.com)

Version: 6 Date: 11 January 2012

Author's response to reviews: see over
Dear Editor-in-Chief

Subject: - Experts comments on Manuscript

Dear Madam,

Reference to the above, Enclosed please find a revised version of our manuscript: “Hepatoprotection with chloroform extract of Launaea procumbens against CCl₄ induced injuries in rat”

This was revised according to the suggestion raised by the reviewers. All points raised by the reviewer were considered in detail. We would like to thank the reviewers for their careful reading of the manuscript and improved copy editing. Their comments improved the quality of the manuscript and we hope that it will now be considered suitable for publication.

Thank you for your attention.

Sincerely yours,

Dr. Rahmat Ali Khan
Assistant Professor
Deptt of Biotechnology,
University of Science and Technology, Bannu